\begin{table}%T2
\caption{\label{table:refero}List of references to Table~\ref{table:oxygen}.
}
%\centering
\par
\begin{tabular}{ll}
 \hline \hline
Abrrev.       & Source                                       \\
\hline
AEL81        &  Alloin  et~al.  (\cite{Alloin1981})     \\
BK02         &  Bresolin \& Kennicutt   (\cite{Bresolin2002})               \\
BKC82        &  Blair et~al.    (\cite{Blair1982})         \\
BKG99        &  Bresolin et~al.   (\cite{Bresolin1999})       \\
C00          &  Castellanos  (\cite{Castellanos2000})                           \\
CPG97        &  Christensen et~al  (\cite{Christensen1997}) \\
DCL88        &  Deharveng  et~al.  (\cite{Deharveng1988})    \\
DK81         &  Dennefeld \& Kunth  (\cite{Dennefeld1981})                   \\
dORW83       &  d'Odorico et~al.   (\cite{Odorico1983})           \\
DTJ80        &  Dufour  et~al.  (\cite{Dufour1980})        \\
DTP87        &  Diaz  et~al.  (1987)        \\
DTV91        &  Diaz  et~al.  (1991) \\
EP84         &  Edmunds  \& Pagel  (\cite{Edmunds1984})                     \\
FGW98        &  Ferguson et~al.  (\cite{Ferguson1998})          \\
FTP86        &  Fierro et~al.  (\cite{Fierro1986})  \\
GOS92        &  Garnett et~al. (\cite{Garnett1992e})         \\
GS87         &  Garnett \& Shields  (\cite{Garnett1987})                   \\
GSP99        &  Garnett  et~al.  (\cite{Garnett1999})    \\
GSS97        &  Garnett  et~al.  (\cite{Garnett1997})    \\
HBC96        &  Henry  et~al.  (\cite{Henry1996})     \\
HPC94        &  Henry et~al.   (\cite{Henry1994})           \\
HPL92        &  Henry  et~al.  (\cite{Henry1992})        \\
KA81         &  Kwitter \& Aller  (\cite{Kwitter1981})                     \\
KG96         &  Kennicutt \& Garnett  (\cite{Kennicutt1996})                 \\
KR94         &  Kinkel \& Rosa  (\cite{Kinkel1994})                       \\
MRS85        &  McCall et~al.  (\cite{McCall1985})            \\
OK93         &  Oey \& Kennicutt  (\cite{Oey1993})                     \\
P70          &  Peimbert  (\cite{Peimbert1970})                              \\
PEB79        &  Pagel  et~al.  (\cite{Pagel1979})     \\
R95          &  Ryder  (\cite{Ryder1995})                                 \\
RPT82        &  Rayo et~al.  (\cite{Rayo1982})    \\
RW97         &  Roy \& Walsh  (\cite{Roy1997})                         \\
S71          &  Searle  (\cite{Searle1971})                                \\
S75          &  Smith  (\cite{Smith1975})                                 \\
SB84         &  Stauffer \& Bothun  (\cite{Stauffer1984})                   \\
SKS96        &  Skillman et~al.   (\cite{Skillman1996}) \\
SS78         &  Shields \& Searle  (\cite{Shields1978})                    \\
SSK91        &  Shields   (\cite{Shields1991})       \\
TPF89        &  Torres-Peimbert   (\cite{Torres1989})  \\
VEP88        &  V\'{\i}lchez et~al. (\cite{lchez1988})       \\
VPD88        &  V\'{\i}lchez  et~al.  (\cite{lchez19881})     \\
vZSH98       &  van Zee  et~al.  (\cite{Zee1998})       \\
WS83         &  Webster \& Smith  (\cite{Webster1983})                     \\
ZKH94        &  Zaritsky  et~al.   (\cite{Zaritsky1994})        \\
 \hline
\end{tabular}
\end{table}